Author’s response to reviews

Title: Follicular dendritic cell sarcoma (FDCS) of urinary bladder with coexisting urothelial carcinoma-A case report

Authors:

Jing Sun (sunjing@ccmu.edu.cn)
Cheng Wang (cheng.wang@nshealth.ca)
Dandan Wang (darlingdan@126.com)
Jiangtao Wu (jiangt_wu@126.com)
Leiming Wang (wangleiming0918@163.com)
Lan Zhao (zhaolan222@163.com)
Lianghong Teng (tenglh2012@163.com)

Version: 2 Date: 26 Aug 2019

Author’s response to reviews:

Dear Editor:

We appreciate the valuable comments by the Reviewers and have now tried to follow their advice. We have modified our manuscript that was significant overlap with other publications, represented authors' names using their full initials in the Authors’ Contributions section, removed any additional files that do not want published alongside my manuscript, improved our language by a native speaker of English. We hope that the manuscript now can be accepted for publication in BMC Urology.

With kind regards

Yours sincerely

Lianghong Teng